Author’s response to reviews

Title: The meaning of working in a person-centred way in nursing homes: a phenomenological-hermeneutical study

Authors:

Tove Karin Vassbø (tove.karin.vassbo@ldh.no)
Marit Kirkevold (marit.kirkevold@medisin.uio.no)
David Edvardsson (D.Edvardsson@latrobe.edu.au)
Karin SJÖGREN (karin.sjogren@umu.se)
Qarin LOOD (Qarin.Lood@neuro.gu.se)
Ådel Bergland (aadel.bergland@ldh.no)

Version: 2 Date: 26 Aug 2019

Author’s response to reviews:


Dear Editor!
Thank you for the thorough second review of our manuscript entitled ‘Thriving at work in nursing homes: meanings of working in a person-centred way (Manuscript ID: NURS-D-19-00123). The changes to the manuscript suggested by the editor have all been appreciated and we have revised the manuscript based on these comments.

Our point-by-point response to editor comments are as follows:

P1, line 1. The title has been changed to ‘The meaning of working in a person-centred way in nursing homes: a phenomenological-hermeneutical study’. The line in person-centred is kept to be consistent with the manuscript.

P3, line 49, the words ‘by organisation researchers’ have been removed.

P5, line 22, the ; has been removed.

P7, line 17, the word ‘staff’s’ has been removed.

P10, line 15, the words ‘Moreover, on’ have been removed.
P12, line 28, the words ‘and six sub-themes’ have been removed.

P13, line 15, the ‘I’ has been removed.

On behalf of the authors,

Best regards,

Tove Karin Vassbø